Author's response to reviews

Title: An increased response to experimental muscle pain is related to psychological status in women with chronic non-traumatic neck-shoulder pain.

Authors:

Anna Sjörs (anna.sjors@liu.se)
Britt Larsson (britt.larsson@liu.se)
Ann L Persson (ap@rct.dk)
Björn Gerdle (bjorn.gerdle@liu.se)

Version: 4 Date: 11 October 2011

Author's response to reviews:

Dear Production team,

We now have made the changes according to your mail of 5th october, i.e.:
- Omitted the footnote
- Removed the blue border beneath the title on the first page
- In the figure legend section we now refer to the figure as figure 1.

With best regards

Björn Gerdle